Author’s response to reviews

Title: Predictors of lower extremity amputation in patients with diabetic foot ulcer: findings from MEDFUN, a multi-center observational study

Authors:

Ejiofor Ugwu (ofornet@yahoo.com)

Olufunmilayo Adeleye (funmiadeleye2004@yahoo.com)

Ibrahim Gezawa (gezawa01@yahoo.com)

Innocent Okpe (innokpe@yahoo.com)

Marcelina Enamino (marcelina.enamino@yahoo.com)

Ignatius Ezeani (ignatiusez@yahoo.com)

Version: 3 Date: 01 Jun 2019

Author’s response to reviews:

Corrections have been effected as follows:

1. The order of the strength/limitations and conclusion sections have been switched such that the conclusion is the last section of the manuscript

2. A full stop has been added in all instances where there is 'et al'.